Author’s response to reviews

Title: An international comparative study of blood pressure in populations of European vs. African descent

Authors:

Richard S Cooper (rcooper@lumc.edu)
Katharina Wolf-Maier (kwolf3@lumc.edu)
Amy Luke (alue@lumc.edu)
Adebowale Adeyemo (deboadeyemo@scientist.com)
Jose R Banegas (joseramon.banegas@uam.es)
Terrence Forrester (terrence.forrester@uwimona.edu.jm)
Simona Giampaoli (simona.giampaoli@iss.it)
Michel Joffres (michel.joffres@dal.ca)
Mika Kastarinen (Mika.Kastarinen@uku.fi)
Paola Primastela (p.primastela@public-health.ucl.ac.uk)
Birgitta Stegmayr (Birgitta.Stegmayr@medicin.umu.se)
Michael Thamm (thamm@rki.de)

Version: 4 Date: 16 November 2004

Author’s response to reviews:

We are sending in a version of the manuscript with a few minor changes. Please use this one instead of the version sent in yesterday.

Thanks-
Richard Cooper, MD